Author's response to reviews

Title: Quality of Diabetes Care Worldwide and the Feasibility of Implementation of the Alphabet Strategy - Global Alphabet Strategy Implementation Audit (GAIA) Project: an international audit of diabetes care, questionnaire, and pilot implementation study

Authors:

James D Lee (douglas.lee@geh.nhs.uk)
Ponnusamy Saravanan (P.saravanan@warwick.ac.uk)
Lakshminarayanan Varadhan (laks.varadhan@gmail.com)
John R Morrissey (john.morrissey70@ntlworld.com)
Vinod Patel (vinod.patel@warwick.ac.uk)

Version: 2 Date: 15 July 2013

Author's response to reviews:

I have include the criteria for authorship, and competing interests in the manuscript as requested.